Author's response to reviews

Title: Ethyl acetate extract of Wedelia chinensis inhibits tert-butyl hydroperoxide-induced damage in PC12 cells and D-galactose-induced neuronal cell loss in mice

Authors:

Wea-Lung Lin (wll@csmu.edu.tw)
Shao-Ming Wang (tu76721@yahoo.com.tw)
Hsing-Chun Kuo (guscsi@gmail.com)
Ying-Jui Ho (yjho@csmu.edu.tw)
Yean-Jang Lee (leeyj@cc.ncue.edu.tw)
Tsui-Hwa Tseng (tht@csmu.edu.tw)

Version: 2 Date: 18 June 2014

Author's response to reviews: see over
Dear Editor:

We have completed the revision of our manuscript (MS: 2028313444132600; Title: Ethyl acetate extract of *Wedelia chinensis* inhibits tert-butyl hydroperoxide-induced damage in PC12 cells and D-galactose-induced neuronal cell loss in mice). Here listed below is the response for the precious comment. Thank you very much for your kind assistance.

Sincerely
Tsui-Hwa Tseng
School of Medical Applied Chemistry
Chung Shan Medical University
No. 110, Section 1, Chien Kuo N. Road
Taichung 402, Taiwan
E-mail: tht@csmu.edu.tw

- Please revise your manuscript to include line and page numbers.

Response:
We have added the line and page numbers in the new manuscript.